Author’s response to reviews

Title: Is annual surveillance of all treated hypothyroid patients necessary?

Authors:

Ananth K Viswanath (aviswanath@nhs.net)
Alison Avenell (a.avenell@abdn.ac.uk)
Sam Philip (sam.philip@arh.grampian.scot.nhs.uk)
Shamsunder H Acharya (S.acharya@nhs.net)
Graeme MacLennan (g.maclennan@abdn.ac.uk)
Katherine Dalziel (kdalziel@nhs.net)
Olivia Pereira (olivia.pereira@nhs.net)
Shirley A Copland (s.copland@nhs.net)
John S Bevan (j.s.bevan@arh.grampian.scot.nhs.uk)
Prakash Abraham (P.Abraham@arh.grampian.scot.nhs.uk)

Version: 5 Date: 25 June 2007

Author’s response to reviews: see over
Dear Editors,

Thank you for accepting the paper for publication in principle. As requested, the formatting changes have been dealt with as detailed below.

**Formatting changes requested:**

**Authors’ contributions** - Please indicate that all authors read and approved the final manuscript.

The statement has been included. The final manuscript has been approved by the authors.

**References** - The reference list should contain all authors’ names, if there are 30 or below. The term *et al.* should not be used.

All authors' names are included in the reference section.

**Insert Tables header.**

Table header inserted.

**Competing interest and funding** - Competing interests.

Changed to competing interests.

I hope the revised manuscripts have addressed all the formatting issues. If you require any further clarification, please feel free to contact us. We look forward to hearing from you.

Yours sincerely

Dr. Viswanath

24/06/07